Author's response to reviews

Title: Neovascular Glaucoma: A Retrospective Review From a Tertiary Center in China

Authors:

Na Liao (nana_jane@163.com)
Chao Hong Li (aoeight@hotmail.com)
Yuan Bo Liang (yuanboliang@gmail.com)
Hui Lv Jiang (hlijiang314@163.com)
Ai Wu Fang (953736995@qq.com)
Sheng Jie Zhou (425760739@qq.com)
Ravi Thomas (ravi.thomas@qei.org.au)
Qin Mei Wang (wqm3@mail.eye.ac.cn)

Version: 2
Date: 19 May 2015

Author's response to reviews:

Dear editors:

Thank you for your letter.

I have revised the manuscript according to your advise (adding the line numbers and acknowledgment) except "2. Requesting for consent statement".

Because the research study is a retrospective study, it is difficult to get consent statement. In order to protect patients' privacy, patient records/information was anonymized and de-identified prior to analysis.

I hope it meets the requirements for publication in BMC ophthalmology. Thank you for your help!

Best wishes!